%!TEX TS-program = xelatex
%!TEX encoding = UTF-8 Unicode



% Style guide
% Keep spelling and punctuation unchanged in quotations
% Regardless of other sources, use People's Climate March
% Use punctuation which is `logical'.
% Use Keystone XL pipeline and Northern Gateway pipeline
% Use \emph{Laudato Si'}
% Use B.C.\ for British Columbia
% Use U.S.\ for United States - except US$
% For periods that don't end sentences use .\
% For sentences that end with capital letters use \@.
% For ellipses use \ldots\
% Use U~of~T
% Use \$1.5 M or B for million or billion
% Use greenhouse gases (with one s)
% Use \href{https://350.org}{350.org}
% Use \href{https://gofossilfree.org}{gofossilfree.org}
% \acronym{UNFCCC} provides small caps for acronyms and initialisms
% \ce{CO2}
% Blockquotes don't need quotation marks, but they do need citations
% Decision-making (adjective) bodies engage in decision making (noun)



% Last edited 2017-03-25



\documentclass[12pt]{article}



% This is the preamble

% Use system fonts
\usepackage{fontspec}
\defaultfontfeatures{Chaparral Pro}

% Define colours for use in this document
\usepackage{color}
\definecolor{dkred}{rgb}{0.9,0.05,0.05}
\definecolor{navy}{rgb}{0.1,0.1,0.4}
\definecolor{gray75}{gray}{0.75}
\definecolor{gray20}{gray}{0.2}

% Indent first graphs
\usepackage{indentfirst}

% Letter sized paper with 1 inch margins
\usepackage[paperwidth=8.5in,paperheight=11in,left=1in,top=1in,right=1in]{geometry}

% Microtype for formatting improvements
% \usepackage[activate={true,nocompatibility},final,tracking=true,kerning=true,spacing=true,factor=1100,stretch=10,shrink=10]{microtype}
% activate={true,nocompatibility} - activate protrusion and expansion
% final - enable microtype; use "draft" to disable
% tracking=true, kerning=true, spacing=true - activate these techniques
% factor=1100 - add 10% to the protrusion amount (default is 1000)
% stretch=10, shrink=10 - reduce stretchability/shrinkability (default is 20/20)

% Package for chemical formulas - use \ce{CO2} for CO2
\usepackage[version=3]{mhchem}

% Define command to put accent in Metis
\newcommand{\Metis}{M\'{e}tis }

% Allow tables
\usepackage{array}
\usepackage{booktabs}

% Bibliography settings
% TK --- Find a way to show dates in footnotes in this style, or make all footnotes render properly in MLA
\usepackage[style=authortitle-icomp,natbib=true,sortcites=true,block=space,maxbibnames=99]{biblatex}
\bibliography{proposal.bib}

% Allow figures
\usepackage{graphicx}
% \usepackage{morefloats}

% Set double spacing
\usepackage{setspace}
\expandafter\def\expandafter\quote\expandafter{\quote\onehalfspacing}
\setstretch{2}

% Use small caps for acronyms and initialisms
\newcommand{\acronym}[1]{\textsc{#1}}

% Code for fancy section headings
\usepackage{titlesec, blindtext, color}
\newcommand{\hsp}{\hspace{20pt}}
\titleformat{\section}[hang]{\Large\bfseries}{\thesection\hsp\textcolor{gray75}{|}\hsp}{0pt}{\Large\bfseries}

% Put commas between adjacent footnotes with \fnsep
% Text.\footnote{Here is a footnote.}\fnsep\footnote{Here is another footnote.}
\newcommand\fnsep{\textsuperscript{,}}

% To allow compact lists
\usepackage{enumitem}

% Make links to figures go to the right place (hopefully!)
% Note - label commands should be flush left to maintain correct pagereferences.
\usepackage{graphicx}
\usepackage[hypcap]{caption}
\usepackage{float}

% Allow URLs
\usepackage{url}
\usepackage[colorlinks=true,allcolors=navy]{hyperref}



\title{Campus Fossil Fuel Divestment Campaign Successes}
\author{Milan Ilnyckyj}



% The preamble is ending now



\begin{document}
	


\maketitle



\href{https://350.org}{350.org} maintains a database of successful divestment campaigns at \href{https://gofossilfree.org}{gofossilfree.org}.
They classify commitments as ``Fossil Free'' (fully divested from the 200 corporations with the largest fossil fuel reserves), ``Full'', ``Partial'', ``Coal and Tar Sands'', and ``Coal only''. 
They also break down organizations by type, including governments, educational institutions, for profit corporations, NGOs, pension funds, philanthropic foundations, etc.\footnote{\url{https://gofossilfree.org/divestment-commitments-classifications/}}
Laval University (listed as ``Full'') is the only Canadian success listed as of February 2017, though a variety of Canadian churches and private foundations have divested.



\vspace{1cm}



In the United States, they list:
\setstretch{1}
\begin{itemize}[noitemsep]
	\item Boston University (Coal and Tar Sands)
	\item Brevard College (Full)
	\item California Institute of the Arts (Full)
	\item Chico State University (Full)
	\item College of the Atlantic (Full)
	\item ESF College Foundation, Inc. (Full)
	\item Foothill-De Anza Community College Foundation (Full)
	\item George School (Coal Only)
	\item Georgetown University (Partial)
	\item Goddard College (Fossil Free)
	\item Green Mountain College (Full)
	\item Hampshire College (Full)
	\item Humboldt State University (Partial)
	\item Naropa University (Full)
	\item Peralta Community College District (Full)
	\item Pitzer College (Full)
	\item Prescott College (Partial)
	\item Rhode Island School of Design (Full)
	\item Salem State University (Full)
	\item San Francisco State University Foundation (Coal and Tar Sands Only)
	\item Stanford University (Coal Only)
	\item Sterling College (Full)
	\item Syracuse University (Full)
	\item The New School (Full)
	\item Unity College (Full)
	\item University of Oregon Foundation (Full)
	\item University of California (Coal and Tar Sands Only)
	\item University of Dayton (Full)
	\item University of Hawaii (Full)
	\item University of Maine System (Coal Only)
	\item University of Maryland (Full)
	\item University of Massachusetts Foundation (Full)
	\item University of Washington (Coal Only)
	\item Warren Wilson College (Full)
	\item Western Oregon University (Partial)
	\item Yale University (Partial)
\end{itemize}
In the United Kingdom they list:\begin{itemize}[noitemsep]
	\item Aston University (Coal and Tar Sands Only)
	\item Birmingham City University (Coal and Tar Sands Only)
	\item Bournemouth University (Full)
	\item Cardiff Metropolitan University (Full)
	\item Cranfield University (Coal and Tar Sands Only)
	\item De Montfort University (Coal and Tar Sands Only)
	\item Goldsmiths University of London (Coal and Tar Sands Only)
	\item Heriot-Watt University (Coal and Tar Sands Only)
	\item King's College London (Coal and Tar Sands Only)
	\item London School of Economics (Coal and Tar Sands Only)
	\item London School of Hygiene \& Tropical Medicine (Coal Only)
	\item Manchester Metropolitan University (Full)
	\item Newcastle University (Full)
	\item Nottingham Trent University (Full)
	\item Oxford Brookes University (Full)
	\item Oxford University (Coal and Tar Sands Only)
	\item Queen Margaret University (Full)
	\item Queen Mary University London (Full)
	\item SOAS, University of London (Full)
	\item Sheffield Hallam University (Fossil Free)
	\item University of Abertay Dundee (Full)
	\item University of Arts Bournemouth (Full)
	\item University of Bedfordshire (Full)
	\item University of Cambridge (Coal and Tar Sands Only)
	\item University of Edinburgh (Coal and Tar Sands Only)
	\item University of Glasgow (Full)
	\item University of Gloucestershire (Coal and Tar Sands Only)
	\item University of Greenwich (Coal and Tar Sands Only)
	\item University of Hertfordshire (Coal and Tar Sands Only)
	\item University of Kent (Full)
	\item University of Lincoln (Full)
	\item University of Portsmouth (Coal and Tar Sands Only)
	\item University of Sheffield (Full)
	\item University of Southampton (Full)
	\item University of St. Andrews (Full)
	\item University of Surrey (Full)
	\item University of Sussex (Coal and Tar Sands Only)
	\item University of Wales Trinity Saint David (Full)
	\item University of Warwick (Full)
	\item University of Westminster (Coal and Tar Sands Only)
	\item University of Worcester (Full)
	\item University of the Arts London (Full)
	\item University of the West of Scotland (Fossil Free)
	\item Wolfson College, Oxford (Coal and Tar Sands Only)
\end{itemize}
\setstretch{2}
As of March 11th, 2017 the list did not include five recently announced divestment commitments: Barnard College and Columbia University, in New York City; the University of Bristol and King's College London in the U.K.; and the National University of Ireland in Galway.\footcite[][]{BarnardDivests2017}\fnsep\footcite[On the process at Barnard, see: ][]{BarnardColumbiaSpec2016}\fnsep\footcite[][]{ColumbiaCoal2017}\fnsep\footcite[On the University of Bristol's website, they announced: ``A key aim is to end investment in companies that derive more than five per cent of turnover from the extraction of thermal coal or oil and gas from tar sands by January 2018''.][]{BristolAnnouncement}\fnsep\footcite[The Kings College decision was report as happening after ``14 days of hunger strike by PhD student Roger Hallam'' and a ``24HR occupation of old Committee room''.][]{KCLHungerStrike}\fnsep\footcite[][]{KCLAgreement}\fnsep\footcite[][]{GalwayDivest2017}




\end{document}